Author's response to reviews

Title: The relationship between baseline Organizational Readiness to Change Assessment subscale scores and implementation of hepatitis prevention services in substance use disorders treatment clinics.

Authors:

Paul W Heideman (paulwheideman@gmail.com)
Hildi J Hagedorn (hildi.hagedorn@va.gov)

Version: 2 Date: 16 October 2009

Author's response to reviews:

The designation of the paper has been changed from short report to research article. Appropriate headings have been added to the abstract.